Author's response to reviews

Title: Alcohol consumption in tertiary education students

Authors:
Nicola J Reavley (nreavley@unimelb.edu.au)
Anthony F Jorm (ajorm@unimelb.edu.au)
Terence V McCann (terence.mccann@vu.edu.au)
Dan I Lubman (dan.lubman@monash.edu)

Version: 3 Date: 28 June 2011

Author's response to reviews:

Dear Reviewers

Thank you very much for your comments.

We have removed the category 'missing' from Tables 2 and 3. In fact, there was no missing data for these questions. We have left the 'Don't know' category as we feel this provides useful information.

Full address details for all the authors have been added to the title page.

Regards,

The authors